**Author's response to reviews**

**Title:** Research in Action: Using Positive Deviance to Improve Quality of Health Care

**Authors:**

Elizabeth H Bradley (elizabeth.bradley@yale.edu)  
Leslie A Curry (leslie.curry@yale.edu)  
Shoba Ramanadhan (shoba.ramanadhan@yale.edu)  
Laura Rowe (laura.rowe@yale.edu)  
Ingrid M Nembhard (ingrid.nembhard@yale.edu)  
Harlan M Krumholz (harlan.krumholz@yale.edu)

**Version:** 4 **Date:** 27 February 2009

**Author's response to reviews:** see over
February 8, 2009

Gregory A. Aarons
Associate Editor
Implementation Science

Dear Dr. Aarons,

Thank you for accepting our paper, #2035701234218268 entitled “Research in Action: Using Positive Deviance to Improve Quality of Health Care” in principle. We are delighted. I have amended the words heart attack to read myocardial infarction as requested. I did not see any other changes requested, but please let me know if I missed something.

Thank you again for your interest in this work. If you have any questions, please contact me at Elizabeth.bradley@yale.edu or 203.785.2937.

Sincerely,

Elizabeth H. Bradley, PhD
Professor of Public Health
Director, Health Management Program
Yale School of Public Health